Author's response to reviews

Title: A stitch in time saves nine? A repeated cross-sectional case study on the implementation of the intersectoral community approach Youth At a Healthy Weight

Authors:

Rianne MJJ van der Kleij (m.j.j.van_der_kleij@lumc.nl)
Mathilde R Crone (m.r.crone@lumc.nl)
Theo GWM Paulussen (theo.paulussen@tno.nl)
Vivian M van de Gaar (v.vandegaar@erasmusmc.nl)
Ria Reis (ria.reis@lumc.nl)

Version: 2 Date: 18 March 2015

Author's response to reviews: see over
Subject: Manuscript revision

17th March 2015

Dear editorial board of BMC Public Health,

Please find attached the revision of our manuscript entitled ‘A stitch in time saves nine? A repeated cross-sectional case study on the implementation of the intersectoral community approach Youth At a Healthy Weight’.

We have added a description of how informed consent was obtained from study participants. “Before participation, informed consent was obtained on audiotape from all participating professionals and transcribed verbatim.”

Furthermore, we have provided details on the ethical approval obtained from the Ethical Committee of the Psychology faculty of the University of Leiden. “This study was approved by the ethical committee of the Faculty of Psychology of the University of Leiden, reference number 8259652117”.

If there are any more questions, please feel free to contact me via email or telephone.

We hope the adjustments are sufficient to be considered for publication in BMC Public Health,

Yours sincerely,

Rianne van der Kleij, on behalf of the authors

Corresponding author: Rianne van der Kleij, Leiden University Medical Center
Department of Public Health and Primary Care, Postbus 9600 zone V-0-P 2300 RC Leiden, The Netherlands. E-mail: m.j.j.van.der.kleij@lumc.nl, Telephone number: 0031-715268406, Fax number: 0031-715268259